Author’s response to reviews

Title: Rubinstein-Taybi syndrome in a Saudi boy with distinct features and variants in both the CREBBP and EP300 genes: A case report

Authors:
Mohammad Al-Qattan (moqattan@hotmail.com)
Abdulaziz Jarman (ajarman@kfshrc.edu.sa)
Atif Rafique (drafiq_2@yahoo.com)
Zuhair Al-Hassnan (zhassnan@kfshrc.edu.sa)
Heba Al-Qattan (heba.qattan96@gmail.com)

Version: 5 Date: 31 Dec 2018

Author’s response to reviews:

"Rubinstein-Taybi syndrome in a Saudi boy with distinct features and variants in both the CREBBP and EP300 genes: A case report" (MGTC-D-18-00379R4)

We have revised the manuscript as follows:

Editor Comments:

1. Please remove the text, ‘The study was approved by the research committee at Riyadh Care (#22-2018)’ from the Title page.
Response: done. See title page

2. Change terms
Throughout your manuscript, please replace the phrase, ‘Mental retardation’ with ‘Intellectual disability.’
Response: Done in abstract/text/Table 1
3. Please remove the heading, ‘Mutation analysis’ and incorporate the corresponding section into the main Case Presentation section.

Response: Done. See Case Presentation

4. Authors’ contributions

The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Please refer to each author by their full initials. Guidance and criteria for authorship can be found in the online journal submission guidelines.

Response: See Authors’ contributions

5. Ethics approval and consent to participate

We note that you have stated in this section that, ‘The father of our patient signed written informed consent to participate in this study.’ In this section, please state whether you have obtained parental consent on behalf of the child included in this study. Please also state whether you had obtained consent to participate from the mother.

Please note that in the case of minors, members of a study are not capable of providing ethical consent to participate for themselves. As research for this manuscript includes participants under the age of 16, we would expect a parent or legal guardian to provide consent to participate in their stead. Please clarify in your Ethics approval and consent to participate subheading whether the parents or legal guardians of these participants provided consent on their behalf.

Response: Done. See Ethics approval and consent to participate

6. Consent for publication

Please remove the text, ‘A copy of the written consent is available for review by the Editor of this journal,’ from this section.

Response: Done. See Consent for publication
7. Remove files

Thank you for providing the CARE checklist, Approval document and Letter from the editor. Please remove the files from the file inventory as they are no longer needed at this stage of the editorial process.

Response: Done

8. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Done

Thanks

M Al-Qattan